Author's response to reviews

Title: Obstetric near-miss and maternal mortality in maternity university hospital, Damascus, Syria: a retrospective study

Authors:

Yara Almerie (yara.mar@hotmail.co.uk)
Muhammad Q Almerie (qutaybah83@hotmail.com)
Hosam E Matar (hematar@hotmail.co.uk)
Yasser Shahrouq (yasseryw@yahoo.com)
Ahmad Abulshamat (drahamd_prof@yahoo.com)
Asmaa Abdulsalam (asma@scs-net.org)

Version: 3 Date: 17 September 2010

Author's response to reviews: see over
Dear Editor-in-Chief,

We gladly received the peer-reviewers comments on our revised manuscript entitled: obstetric near-miss and maternal mortality in maternity university hospital, Damascus, Syria: a retrospective study.

As kindly advised by yourself, we undertook a thorough copy-editing of our manuscript and are pleased to submit our revised version to your consideration.

We are certain that this piece of work will help policy-makers not only in Syria but also in other countries facing similar challenges. It will hopefully serve as a guiding tool which helps in directing expenditure of the modest national health budgets towards maternal health care priorities.

Should you have any queries, please do not hesitate to contact me.

Kindest regards,

Muhammad Q Almerie